Author's response to reviews

Title: Factors associated with access to HIV care services in eastern Uganda; the Kumi home based HIV counseling and testing programme experience

Authors:

David Lubogo Dr. (Lubogodad@yahoo.com)
John B Ddamulira Dr. (jddamulira@musph.ac.ug)
Raymond Tweheyo Dr. (rtweheyo@musph.ac.ug)
Henry Wamani Dr. (hwamani@musph.ac.ug)

Version: 8 Date: 28 August 2015

Author's response to reviews: see over
28th August, 2015

Dr. Elaine Zhang
Executive Editor
BMC Family Practice
BioMed Central
236 Gray's Inn Road
London WC1X 8HB
United Kingdom

Dear Dr. Elaine Zhang,

Re: Submission of our revised manuscript entitled ‘Factors associated with access to HIV care services in eastern Uganda; the Kumi home based HIV counseling and testing programme experience’ in BMC Family Practice.

Please receive our revised manuscript entitled as above for your kind attention. In this revised manuscript, we have attempted to address all the reviewers' comments as described in detail in the Response to Reviewer Comments. We have also made the necessary changes to the format of the paper as per the BMC Family Practice Instructions for Authors.

Sincerely,

Dr. David Lubogo MBChB, MPH
Department of Community Health and Behavioural Sciences
Makerere University School of Public Health
Tel: +256 772 362372
Email: dlbogo@musph.ac.ug